Author's response to reviews

Title: Mitral paravalvular abscess with left ventriculo-atrial fistula in a patient on dialysis

Authors:

Tadashi Kitamura (funcorogash@hotmail.com)
James Edwards (james.edwards@health.sa.gov.au)
Suchi Khurana (suchi.khurana@health.sa.gov.au)
Robert G Stuklis (robert.stuklis@health.sa.gov.au)

Version: 2 Date: 9 July 2009

Author's response to reviews:

9th July 2009

Mr Vipin Zamvar
Editor-in-Chief
Journal of Cardiothoracic Surgery
Edinburgh
United Kingdom

Dear Mr Zamvar

Many thanks for accepting our manuscript entitled “Mitral paravalvular abscess with left ventriculo-atrial fistula in a patient on dialysis” for publication in Journal of Cardiothoracic Surgery.

Please find the attachment in which I have made required changes;
1. Addition of Consent section.
2. Changing “Figure 3A” and “Figure 3B” into “Figure 3” and “Figure 4” respectively, and amendment of Figure legends accordingly.
3. Deletion of the words “authors” and “title” from the title page.

Sincerely,

Tadashi Kitamura MD, PhD
Cardiothoracic Surgical Unit
Royal Adelaide Hospital
North Terrace
Adelaide SA 5000
Australia